## **REVIEW HISTORY**

Paper: "Caspian Sea Convention: the settlement of the long-lasting dispute and

its implications"

Corresponding Author: Temirtay Toktassynov

Email: ttoktassynov@yandex.com

Doi: 10.19044/elp.v6no3a1

Peer review:

Reviewer 1: Blinded

Reviewer 2: Blinded

Published: 30.09.2019

